Author's response to reviews

Title: Validation of the Spanish version of the Oral Health Impact Profile (OHIP-14Sp) in elderly Chileans.

Authors:

Soraya León (sleon@utalca.cl)
Daniel Bravo-Cavicchioli (dbravo@utalca.cl)
Gloria Correa-Beltrán (gcorrea@utalca.cl)
Rodrigo A Giacaman (giacaman@utalca.cl)

Version: 3
Date: 27 May 2014

Author's response to reviews: see over
Cover Letter

**Title:** Validation of the Spanish version of the Oral Health Impact Profile (OHIP-14Sp) in elderly Chileans.

**Authors:** Soraya León, Daniel Bravo-Cavicchioli, Rodrigo A. Giacaman, Gloria Correa

Drs. Mendes, Morgan, Robinson, Sculean and Senpuku
Section Editors
BMC Oral Health

After the first submission there were some editorial requests that were addressed in the present submission. A list of the modifications follows:

Editorial request:

1. **Abstract.** Abstract has been uploaded and it is consistent in both the manuscript and the submission website.

2. **Acknowledgements.** This section has been included in the manuscript, according to the instructions.

3. **Line Numbering.** Line and page numbers are now included.

4. **Tables.** Tables have been rearranged according to the journal instructions.

Yours sincerely,

Rodrigo A. Giacaman, DDS, PhD.
Corresponding author